 
CENWP-OP-B                                                                                                    1 February 2000

MEMORANDUM FOR CENWP-OP, ATTN:  Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 4 of 2000, which covers the period 23 through 29 January.
1.  OPERATION SUMMARY.

1.1. Daily average river flow ranged from 177.0 to 232.1 kcfs.  Daily average tailwater elevation for the project ranged from 16.4’ to 18.6’ msl.  River temperature ranged from 40 to 41° F.  Spill bays 1 and 18, which supply attraction flow for the Cascades Island ladder and B-branch ladder, respectively, remain on sill. 

1.2.  Unit 10 remained out of service for stator rewind.  Units 1, 2, and 3 were out of service from approximately 0700-1700 on 23 through 28 January for divers to install surface collector equipment.  Unit 17 was out of service from 0924 on 26 January through 1342 on 27 January for vacuum breaker repairs.  

1.3.  The first powerhouse ice and trash sluiceway remained open for debris removal.  The second powerhouse sluiceway remained closed this reporting week. 
1.4.  Second powerhouse fish units remained in service.  The units are being placed on standby from approximately 0000-0300 if drawdown exceeds 1.0’.
2.  ACTIVITIES. 
2.1.  Fish Ladder and DSM Service.  The Bradford Island ladder including A and B-branches, PH1 DSM, and PH2 DSM and flume remained out of service for winter maintenance.  

2.2. AWS Closures.  There were no AWS closures this reporting week.   

2.3. Dewaterings and Fish Salvage.  There were no dewaterings or fish salvage this reporting week.       

3.  RESEARCH. 
3.1. Normandeau MGR Testing.  Normandeau and Mid-Columbia Consultants completed biological MGR testing at units 5 and 6 on 29 January. 

3.2. WDFW Fish Counting.  WDFW continued 24-h/d video counting at Washington shore. 

4.  FISHWAYS.  Project biologists inspected on 24, 25 and 28 January this reporting week.

4.1. Fish Passage Plan Observations. 

DATE
LOCATion
FPP Deviation
CAUSE
RESPONSE

1/28
PH2 collection channel velocity meter.
Meter was not working and should register 1.5-4.0 fps.
Unknown.
No action taken.

4.2. Sensor Calibration.  Water level sensors were checked for calibration on 26 January this reporting week.   

4.3. Zebra Mussel Monitoring.  The zebra mussel monitoring station was inspected and showed no signs of colonization this reporting week.

5. CONSTRUCTION UPDATE.
5.1. The PH2 DSM and flume will be watered-up on 3 February.

                                                                                                     DEBORAH L. CHENOWETH

                                 

                                                              Operations Project Manager
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